Author’s response to reviews

Title: Positive surgical margins and biochemical recurrence following minimally-invasive radical prostatectomy - An analysis of outcomes from a UK tertiary referral centre

Authors:

Ashwin Sachdeva (ashwin.sachdeva@gmail.com)
Rajan Veeratterapillay (Rajan.Veeratterapillay@nuth.nhs.uk)
Antonia Voysey (antonia.voysey@gmail.com)
Katherine Kelly (kathykelly2490@gmail.com)
Mark Johnson (Mark.Johnson@nuth.nhs.uk)
Jonathan Aning (Jonathan.Aning@nuth.nhs.uk)
Naeem Soomro (naeem.soomro@nuth.nhs.uk)

Version: 2  Date: 23 Aug 2017

Author’s response to reviews:

Dear Dr Henderson,

We would like to thank the reviewers for their comments on the revised manuscript.

As advised, we have now made the recommended editorial revisions to the manuscript and hope that it is now acceptable for publication in BMC Urology.

All changes have been highlighted in yellow.

Yours sincerely

Ashwin Sachdeva